ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Pula, Gabor; Peltonen, Tuomas A.

Working Paper Has emerging Asia decoupled? An analysis of production and trade linkages using the Asian international input-output table

ECB Working Paper, No. 993

Provided in Cooperation with: European Central Bank (ECB)

Suggested Citation: Pula, Gabor; Peltonen, Tuomas A. (2009) : Has emerging Asia decoupled? An analysis of production and trade linkages using the Asian international input-output table, ECB Working Paper, No. 993, European Central Bank (ECB), Frankfurt a. M.

This Version is available at: https://hdl.handle.net/10419/153427

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



WORKING PAPER SERIES NO 993 / JANUARY 2009

B EZB EKT EKP

HAS EMERGING ASIA DECOUPLED?

AN ANALYSIS OF PRODUCTION AND TRADE LINKAGES USING THE ASIAN INTERNATIONAL INPUT-OUTPUT TABLE

by Gabor Pula and Tuomas A. Peltonen





NO 993 / JANUARY 2009

HAS EMERGING ASIA DECOUPLED?

AN ANALYSIS OF PRODUCTION AND TRADE LINKAGES USING THE ASIAN INTERNATIONAL INPUT-OUTPUT TABLE'

by Gabor Pula and Tuomas A. Peltonen²

This paper can be downloaded without charge from http://www.ecb.europa.eu or from the Social Science Research Network electronic library at http://ssrn.com/abstract_id=1311786.



In 2009 all ECB publications feature a motif taken from the €200 banknote.



ZB EKT EKC

 The views presented in this paper are those of the authors and do not necessarily reflect those of the European Central Bank (ECB). We would like to thank Ettore Dorrucci, Marcel Fratzscher, Frank Moss, Daniel Santabarbara, Roland Straub, Christian Thimann, and other colleagues at the International Policy Analysis Division of the European Central Bank for fruitful discussions at different stages of this project. Thanks also to seminar participants at the ECB Globalization, trade and competitiveness forum, especially to Robert Anderton, as well as to conference participants at the 16th Central Banking Seminar, Seoul, October 21-24, 2008.
 Correspondence should be sent to: European Central Bank, Kaiserstrasse 29, D-60311 Frankfurt am Main, Germany;

e-mail: gabor.pula@ecb.europa.eu; tuomas.peltonen@ecb.europa.eu

© European Central Bank, 2009

Address Kaiserstrasse 29 60311 Frankfurt am Main, Germany

Postfach 16 03 19 60066 Frankfurt am Main, Germany

Telephone +49 69 1344 0

Website http://www.ecb.europa.eu

Fax +49 69 1344 6000

All rights reserved.

Any reproduction, publication and reprint in the form of a different publication, whether printed or produced electronically, in whole or in part, is permitted only with the explicit written authorisation of the ECB or the author(s).

The views expressed in this paper do not necessarily reflect those of the European Central Bank.

The statement of purpose for the ECB Working Paper Series is available from the ECB website, http://www.ecb.europa. eu/pub/scientific/wps/date/html/index. en.html

ISSN 1725-2806 (online)

CONTENTS

Abstract	4
Non-technical summary	5
1 Introduction	7
2 Brief review of the literature	9
3 Emerging Asia's external dependence based on trade data	п
4 Methodology	14
 5 Analysis using the AIO tables 5.1 "Backward linkages" of production 5.2 Contribution ratios of final demand 	6 6
to value added	19
6 Conclusions	23
References	25
Appendices	27
European Central Bank Working Paper Series	40

Abstract

Due to the emergence of global production networks, trade statistics have became less accurate in describing the dependence of emerging Asia on external demand. This paper analyses, using an update of the Asian International Input-Output (AIO) table, the interdependence of emerging Asian countries, the United States, the EU15, and Japan via trade and production linkages. According to the results, we do not find evidence of the decoupling of emerging Asia from the rest of the world. On the contrary, we find evidence on increasing trade integration, both globally and regionally. Nonetheless, our analysis indicates that emerging Asia's dependence on exports is only about one-third of its GDP, i.e. well below the 50% exposure suggested by trade data. This finding can be explained by the high import content of exports in these economies, which is a result of the increasing segmentation of production across the region.

Keywords: emerging Asia, Asian International Input-Output table, real linkages, decoupling, resilience

JEL classification: F14, C67, E23.

Non-technical summary

Related to the ongoing debate on "decoupling" of emerging markets, the paper analyses, using an update of the Asian International Input-Output (AIO) table, the dependence of emerging Asia¹ on demand from the region itself, from the advanced economies, in particular Japan, the United States, and the EU15 (the G3 countries henceforth), and from the rest of the world. Due to the emergence of global production networks, trade data has became less accurate in describing the interdependences of the economies in emerging Asia. There are two main shortcomings of trade data. First, trade statistics are unable to capture the source of valueadded (i.e. to quantify the contribution of each country to the total value added produced) in the production chain. Thus, trade statistics provide inaccurate information on the exposure of each country in the production chain. Second, because trade data are gross statistics they are prone to double-counting. The more the production is segmented across countries the higher the total volume of trade will be, and thus, the more trade data overestimate the openness of emerging Asia as a region.

These problems can be overcome by using an international input-output table to analyze the trade and production linkages between countries and sectors. The Asian International Input-Output table by the Institute of Developing Economies Japan External Trade Organization (IDE-JETRO) provides detailed information on trade and production linkages between 9 economies in the Asia-Pacific region: China, Indonesia, Malaysia, the Philippines, Singapore, South Korea, Taiwan (Republic of China), Thailand, and Japan, as well as the United States. The geographical breakdown for trade also includes Hong Kong S.A.R., the EU15 and the rest of the world. To date, the AIO tables have been compiled for the years 1985, 1990, 1995 and 2000. Given the rapid changes in production and trade structures, however, these data look inadequate to describe the current situation. For this reason, we calculate the country-level update of the AIO table for 2006.

The updated 2006 AIO table is used for two analyses. First, we calculate the linkages via imported inputs (the so-called "backward linkages" of production), which allows us to describe the inter-linkages of the emerging Asian countries *in the production process*. Second, as the main contribution of the paper, the reliance of each country's value added on *domestic demand, intra-regional and extra-regional demand* is computed. Comparing results from the 1995, 2000, and the updated 2006 AIO tables we also report the evolution of major trends in trade and production in the region.

¹ Emerging Asia in our analysis contains China, Hong Kong S.A.R., Indonesia, Malaysia, the Philippines, Singapore, South Korea, Taiwan (R.O.C.), and Thailand. Other countries of emerging Asia are not included due to data limitations.

The main findings of the paper are the following. First, only about one-third of the value added in emerging Asian countries is determined by external demand, significantly lower than the 50% exposure suggested by the aggregate trade data, while domestic demand contributes around two-thirds to the value added. Second, the dependence of emerging Asia's value added on export markets has steadily risen since 1995, a phenomenon in line with increasing global trade integration, and a clear evidence against the decoupling view. Third, although intraregional and Chinese markets have both gained importance, they still account for only around 7% of the final demand. This share is also below the one suggested by trade data. As regards extra-regional markets, the G3 economies accounted for 16% of total final demand in 2006, with an increasing dependence of emerging Asia on the EU15, and a declining importance of US and Japanese markets. Moreover, demand from the rest of the world has recently grown substantially, accounting for 14% of total final demand in 2006 – a share equal to that of the G3 countries.

As it is evident from these results, the paper finds no support for the decoupling view. At the same time, however, it finds that, if the bias in trade data due to the segmentation of production is accounted for, the exposure of emerging Asia to external demand is significantly lower than suggested by trade statistics. In other words, on the one hand we find no evidence of decoupling, but on the other hand we calculate that emerging Asia is less "coupled" with the rest of the world than suggested by trade data.

When interpreting the results, one should note the caveat that the analysis with the AIO table can only capture the direct trade effects, i.e. neither any "second-round" effects of an export slowdown on domestic demand via lower employment, wages or investment, nor any financial market or policy related channels are accounted for.

Our findings on the production structure of the Asian hub and the role of China within it, also add some interesting insights to the literature. The "backward linkages" of production indicate a changing role of China in the Asian hub. Rather than being a last stage assembler, China increasingly takes over the role of Japan and supplies inputs for the production in other countries of the region. This finding is in line with the changing structure of global production networks and the downsizing of manufacturing activities in advanced economies.

1. INTRODUCTION

Since 1998, emerging Asia's² exports more than doubled in value, an increase well above the growth rate of overall world demand. As a result, the share of emerging Asia's exports in total world exports increased from 17% in 1998 to 22% in 2007. Moreover, by accounting for 14% of the world GDP³ and contributing nearly half of world's GDP growth⁴, emerging Asia has become a key to world's economic growth and dynamics.

The slowdown of the US economy since the second half of 2007 and the continued strength of growth in emerging Asia have set off, the so-called "decoupling" debate on whether emerging Asia has decoupled from the global business cycle. In general, decoupling can be defined as "the emergence of a business cycle dynamic that is relatively independent of global demand trends and that is driven mainly by autonomous changes in internal demand" (ADB (2007)). We use this definition of decoupling in the analysis of this paper.

At the early stages of the recent financial crisis, the shocks hitting the global economy seemed to be primarily US-based, emanating from the collapse of the US housing bubble. Since early 2008, however, a broader set of shocks has appeared, including a global banking and liquidity crisis with negative implications on financing costs, risk premia, and availability of credit, various commodity price shocks, and emergence of housing-related problems in several non-US economies. Thus, the relevant question today is more to what extent emerging Asia has decoupled not only from the United States, but from extra-regional demand in general. For this reason, our analysis focuses on the extra vs. intra-regional determinants of economic growth in emerging Asia.

The main arguments behind the decoupling theory are threefold. First, according to the trade statistics, trade linkages of emerging Asia with the G3 countries⁵, and in particular with the United States, are less important today than in the past. Indeed, the steady slowdown of exports to the United States since 2006 has been compensated by dynamically expanding export markets to other emerging economies. Second, prolonged productivity and income

² In general, emerging Asia is defined in the paper as consisting of China, Hong Kong S.A.R., Indonesia, Korea, Malaysia, the Philippines, Singapore, Taiwan (R.O.C.), and Thailand. However, in the trade analysis using UN COMTRADE data (Chapter 2), Taiwan (R.O.C.) is not included due to missing data. In contrast, in the analyses with the Asian International Input-Output (AIO) tables, emerging Asia does not include Hong Kong due to the fact that this country is not included in the production matrix of the AIO table.

³ Measure in USD 2007 values.

⁴ Based on the IMF World Economic Outlook April 2008 projections for 2008 using PPP weights.

⁵ By the G3 economies, we mean the United States, the euro area (EU15), and Japan.

growth, as well as rising purchasing power are increasing emerging Asia's own final demand. A major reason, why emerging Asia's business cycle may have decoupled, is China's emergence with a domestic market of 1.3 billion consumers. In the view of the supporters of the decoupling theory, China is an engine of growth in emerging Asia, i.e. it increasingly demands for goods produced in other countries of the region. Finally, unlike in the earlier episodes of a global slowdown, emerging Asian countries are now better able to weather adverse external conditions by active economic policies. This is because most countries now have current account surpluses, large foreign reserves and many even budget surpluses, leaving room for a fiscal stimulus.

The paper aims at analyzing the dependence of emerging Asia through trade linkages on the demand both from the region itself and on the advanced economies, especially the United States, the EU15 and Japan. There are three main questions that we intend to answer.

- 1. To what extent value added in emerging Asia is determined by domestic versus external demand? Has the dependence on external factors decreased over time? (In the latter question, "yes" supports the decoupling view).
- 2. How important is intra-regional trade in emerging Asia? Has the increasing purchasing power in China and other emerging Asian countries provided an expanding market for products from the region, helping to isolate emerging Asia from global business cycle fluctuations? ("yes" for the latter supports the decoupling view)
- 3. To what extent value added in emerging Asia depends on demand from US, Japan and the euro area and the rest of the world? Has the relative importance of these regions changed?

Due to the emergence of global production networks, it is less accurate to analyze economic dependences between countries by using only trade data. The main shortcoming of trade data lies with its inability to capture the source of value-added, i.e. to quantify the contribution of each country to the total value added produced in the production chain. This can be overcome by using an international input-output table to analyze the real linkages between countries and sectors.

One should note, however, there are some limitations to this approach. The analysis with the AIO table can only capture the "direct" trade effects, i.e. neither any "second-round" effects of an export slowdown on domestic demand via e.g. lower employment, wages or investment, nor any financial or policy related channels cannot be taken into account. Thus, the actual impacts of negative external demand shock may be underestimated by the numbers provided in the paper. The paper also lacks a sector level analysis. Although the 1995 and 2000 AIO tables provide information on sectors, the update of the AIO table at the sectoral level is currently not possible due to severe data limitations. However, an extensive literature

of sectoral studies is available (see e.g. Dieter (2007), Gangnes and Van Assche (2008), Nag et. al (2008), Luthje (2004)), which can be used to supplement the findings of this paper.

Compared to the study closest to our work, Mori and Sasaki (2007), this paper contains at least three improvements. First, as regards the updating procedure of the AIO table, the paper takes into account the compositional shift in the imports from final to intermediate goods, and also applies an adjustment for Hong Kong's entrepot trade. Second, the paper gives a picture of production linkages in the region using the Leontieff coefficients of the AIO table. Finally, the paper presents a broad set of descriptive results on income dependency, with specific attention paid to the reliance of emerging Asia's GDP on European markets.

The paper has some interesting contributions to the decoupling debate. We find no evidence of decoupling of the emerging Asian region from the rest of the world. On the contrary, we find evidence on increasing trade integration, both globally and regionally. Our results indicate that emerging Asia's GDP is increasingly driven by exports, in line with the stronger economic integration of the world economy. Intra-regional markets, despite of gaining substantially in importance in the recent years, still account for only 7% of value added in the region. Nevertheless, the paper finds that domestic demand, with a share of around two thirds of the final demand, is still key to the economic growth in the region. Consequently, the share of external demand of around one third is, therefore, significantly lower than the 50% exposure suggested by aggregate trade data.

The paper is organized as follows. Section 2 briefly discusses the related literature. Section 3 introduces some stylized facts based on trade statistics and describes the limitations of these data. Section 4 presents the methodology used to update the AIO table, and Section 5 the main analysis. Finally, Section 6 concludes. Technical details on the structure of the AIO table, the updating procedure, and the derivation of measures used in the analysis are presented in the Appendix.

2. BRIEF REVIEW OF THE LITERATURE

The existing empirical evidence on the decoupling of emerging Asia is ambiguous. In support of the decoupling view, several recent studies suggest that global (common) factors play a relatively less important role in driving business cycles in emerging Asia than in other regions of the world (see e.g. IMF (2007), and Dees and Vansteenkiste (2007)). Moreover, the importance of common factors seems to have declined since the mid-eighties. In parallel, the studies indicate an increasing role of regional factors, in line with ongoing trade and financial integration in emerging Asia (see ADB (2008)).

In contrast, several studies indicate an increasing synchronization of business cycles of advanced and emerging Asian economies. According to calculation of IMF (2007) and ADB (2008), import demand from the United States, the euro area, and Japan is assumed to be now more important for the region than ever before. Indeed, estimations suggest that the co-movements between (non-oil) import demand from the aforementioned G3 economies and economic growth of emerging Asia became stronger in the last decade compared to earlier periods.⁶

Moreover, evidence based on existing trade data does not support the view that intra-regional demand for final goods is increasing, and that China is emerging as an engine of growth for the region. Although intra-regional exports are increasing fast, it is mostly due to trade in intermediate goods. In fact, there is no indication that exports of final goods from emerging Asia to China, or to other countries of the region would have risen strongly (ADB 2008).

One should note, however, that decoupling does not mean that a slowdown of the growth in the United States or the global economy would not have an impact on growth in emerging Asian economies. It means that the GDP growth in these countries will slow by much less than in previous recession episodes. Combining elasticities from panel estimates⁷ and actual trade data, IMF (2007) calculates that the impact of a 1% slowdown in the US GDP growth has a -0.15 percentage point (pp) impact on growth in emerging Asia. The finding that a 1% slowdown in the euro area would have an impact closely similar in size implies that a broader slowdown within the G3 group can have an economically significant effect on emerging Asia's growth. However, these elasticities, given that they do not allow for spillovers between countries, are assumed to be underestimated. Indeed, VAR estimates that allow for intercountry dependencies estimate the impact of a 1% slowdown in the US GDP growth around -0.4 pp ((IMF 2007) and ADB (2008)). Moreover, Dees and Vansteenkiste (2007) using a global VAR model estimate the impact in the range of -0.16 to -0.30 pp⁸. Macro-model simulations that also take into account changes in relative prices and allow for a depreciation of the US dollar indicate impacts in the range of -0.5 to -1.5 pp (IMF (2007, ADB (2008)). Finally, when simulations take into account factors, such as cross-country inter-linkages in business and consumer confidences, integration of financial markets and synchronization of policy decisions, the elasticities may easily exceed those cited above.

 $^{^{6}}$ According to the IMF estimates, the rise in the openness of emerging Asia from 4.8% in 1981-85 to 7.1% in 2001-2005 (measured as merchandise exports to GDP) resulted in an increase of elasticity of growth to US growth by 0.2 percentage points.

⁷ The sample contains 130 countries and data from 1970-2005. For more details see IMF (2007) p. 132.

⁸ Including "echo effects" via trade links between third countries, the authors estimate an impact in the range of 0.2-0.4 percentage points.

In a study methodologically close to our paper, Mori and Sasaki (2007) use the updated version of the AIO table to quantify interdependences in the Asia-Pacific region. According to their results, interdependencies between the Asia-Pacific economies in terms of global production networks deepened further in 2000-2005, while China became the main production center in the region. The authors also find that the East Asian economies, rather being more autonomous, became more exposed to economic developments outside of the region.

3. EMERGING ASIA'S EXTERNAL DEPENDENCE BASED ON TRADE DATA

This section briefly summarizes the stylized facts on emerging Asia's external dependence based on trade statistics. The trade data we use is from UN COMTRADE, and it only contains data on traded goods for 1998-2006.

First we start the analysis with aggregated exports data. The main findings are the following (see Table 1):

- Exports (in goods) contributed 45% of GDP in 2006, which indicates a strong exposure of the region to external demand. Moreover, the exports to GDP ratio increased significantly from 34% in 1998 to 45% in 2006, giving no support to the decoupling theory. Using the more complete National Accounts statistics, which include trade in both goods and services, and serve as a benchmark for the final findings of the paper, exports of emerging Asia accounted for 53% of GDP in 2006. As for comparison, in 2006, the exports-to-GDP ratio in the United States was 11%, in the EU15 16%, and in Japan 16%.
- Intra-regional demand determines 17% of GDP. The role of intra-regional market has increased, mainly driven by a robust expansion of the Chinese market. The contribution of exports to China in the total value added increased from 6% to 12% in 1998-2006.
- As regards extra-regional demand, exports to the G3 countries accounts for 19% of GDP, slightly up from 16% in 1998. The US markets are the most important (8%), followed by the EU15 (7%), and Japan (4%). Demand from the rest of the world determined 10% of total value added in 2006.

			In millio	on USD					in % o	f GDP		
	Т	otal	Fi	nal	Interr	nediate	Тс	otal	Fi	nal	Intern	nediate
	1998	2006	1998	2006	1998	2006	1998	2006	1998	2006	1998	2006
Exports to G3 countries	363.0	905.1	143.0	289.2	220.0	615.9	16.2	18.7	6.4	6.0	9.8	12.7
within that to US	164.5	378.6	70.8	138.8	93.7	239.8	7.3	7.8	3.2	2.9	4.2	4.9
to the EU	114.7	327.1	39.5	96.3	75.2	230.9	5.1	6.7	1.8	2.0	3.4	4.8
to Japan	83.9	199.4	32.8	54.2	51.0	145.2	3.7	4.1	1.5	1.1	2.3	3.0
Intra-regional exports	245.4	800.9	45.7	84.5	199.7	716.4	11.0	16.5	2.0	1.7	8.9	14.8
within that exports to China*	78.3	271.9	9.4	13.0	68.9	258.8	6.2	12.1	0.8	0.6	5.5	11.5
RoW	160.6	492.2	47.1	121.2	113.6	370.9	7.2	10.2	2.1	2.5	5.1	7.7
Exports of goods, total	769.1	2198.2	235.8	495.0	533.2	1703.2	34.4	45.3	10.5	10.2	23.8	35.1
Memo item Exports, goods and services	958.9	2588.3					42.8	53.4				

Table 1. Exports by type of goods and by destination

Note: The numbers refer to exports of goods. * GDP ratio is based on non-China emerging Asia GDP. Source: UN COMTRADE database.

Aggregated (total) exports data indicate that emerging Asian countries are relatively open, are increasingly integrated in global trade networks and increasingly dependent on external markets. In a word, aggregated trade data reject the decoupling theory. Nevertheless, when a more detailed dataset is used and the analysis includes exports by types of goods, the picture becomes more ambiguous.

Intermediate goods exports accounted for 77% of total exports in 2006. Moreover, exports of intermediate goods have expanded more dynamically than exports of final goods, and have been the main contributor to increasing openness of emerging Asia in most country-relations. However, one may argue that trade of intermediate goods, being a result of production segmentation and prone to double-counting, should be excluded from the analysis. When calculating final demand dependence, it is exports of final goods that matter.

According to final goods exports, openness of emerging Asia is low and has not increased over the last decade (10.2% in 2006 vs. 10.5% in 1998). Extra-regional demand contributed to GDP by 8.5% in 2006, similarly to 1998. However, the importance of US and Japanese markets have declined, while the importance of EU and rest of the world has increased. *Unlike data on total exports, final goods trade statistics do not justify the high and increasing exposure of emerging Asia to external demand*.

Which data should we trust? Assuming that production networks aim at supplying export markets, and thus intermediate goods should partly be included in the analysis, we believe the actual exposure of emerging Asia should lie somewhere between the numbers suggested by total and final goods export data. However, trade statistics itself do not provide enough information to tell the exact exposure.

With the emergence of global production networks, trade data has became less accurate in describing the interdependences of the economies in emerging Asia. There are two main shortcomings of trade data. First, trade statistics are unable to capture the source of value-added (i.e. to quantify the contribution of each country to the total value added produced) in the production chain. Thus, trade statistics provide inaccurate information about the dependence of each country in the production chain on external demand. Second, because trade data are gross statistics they are prone to double-counting. The more the production is segmented across countries the higher the total volume of trade will be, and thus, the more trade data overestimate the openness of emerging Asia as a region. Let us illustrate these problems with a numerical example.

Assume that the production chain contains three countries: Malaysia supplies China with intermediate inputs, China uses these inputs for both producing final goods to its domestic market and to exports to EU markets. In order to calculate the impact of changes in demand from the EU on the value-added of Malaysia and China, one needs to know the share of inputs from Malaysia in the value of final goods produced by China to its domestic and export markets. This information, however, is not provided by the trade data.

Figure 1: Sources of value added in the production chain, an illustrative example

intermediate goods from Malaysia= 2 bn USD



	Malaysia to China	Malaysia to EU	China to EU
Actual exposure	2	8	42 (=50-8)
Trade data (total exports)	10	0	50
Trade data (final goods exports)	0	0	50

Assume that the value of inputs exported from Malaysia to China is USD 10 bn, and the value of final goods exports from China to the EU15 is USD 50 bn⁹ (see Figure 1). Assuming that USD 2 bn of inputs from Malaysia ends up in products that are consumed in China, the products exported to the EU15 will contain USD 8 bn value-added from Malaysia and USD 42 bn value-added from China. Consequently, fall in the demand of EU15 by 50 bn US

⁹ The illustrative numbers closely track actual export values in 2006, by both types of goods and country relations.

dollars would have an 8 bn impact on Malaysia and a 42 bn impact on China. Aggregate trade data (similar to trade data on final goods), however, would indicate a 50 bn USD impact on China and no impact on Malaysia.

The above example can also be used to illustrate the problem of double-counting. Malaysian inputs that end up in Chinese exports to the EU are counted twice: once when they are exported from Malaysia to China and once when they are exported from China to the EU. As a result, while the actual value-added that is exported from the region, is USD 50 bn, trade data would indicate an USD 60 bn of total exports of the region as a whole.

A possible way to compass these problems is to use input-output tables. Input-output tables are built on a broad set of disaggregated statistics and take into account not only trade flows, but also information on flows of inputs within the production process. In the following analysis, we will use the Asian International Input-Output tables to describe the dependence of emerging Asian countries on intra- and extra-regional demand.

4. METHODOLOGY

This Section describes the structure of the Asian International Input-Output (AIO), the updating methodology, as well as the sensitivity analysis applied.

The Asian International Input-Output (AIO) tables are compiled by the Institute of Developing Economies Japan External Trade Organization (IDE-JETRO), and can be used to analyze the structures of industry and trade linkages, as well as inter-temporal changes in the interdependences of the economies in the Asia-Pacific region.

The AIO tables provide detailed information on trade and production linkages between 9 economies in the Asia-Pacific region: China, Indonesia, Malaysia, the Philippines, Singapore, South Korea, Taiwan (R.O.C.), Thailand, and Japan as well as the United States. The geographical breakdown for trade also includes Hong Kong S.A.R., the EU, and the rest of the world. The AIO tables contain the input-output tables of these countries linked together using detailed trade matrices. Accordingly, the AIO tables have both a country and a sectoral dimension, which makes it possible to describe inter-linkages between various sectors of different countries.¹⁰ A detailed description of the structure of the 2000 AIO table is given in Appendix A.1.

¹⁰ For more technical details on the AIO tables, see Asian International Input-Output Table 2000, Volume 1. Explanatory Notes, JETRO, March 2006

To date, the AIO tables have been compiled for the years 1985, 1990, 1995 and 2000. Given the rapid changes in production and trade structures, however, these data look inadequate to describe the current situation. For this reason, we update the AIO table at the country level for 2006, and use it to analyze the research questions of the paper. The updating procedure is described in detail in Appendix A.2.

To test the validity of the updating procedure, we did the following two sensitivity analyses. First, we examined the residuals from the GDP identity. Input-output tables are closed systems, meaning that they are constructed so that total demand equals total supply. Given that we use imports data to update the trade linkages in the input-output table, any discrepancy between imports data reported by the importer and export data reported by the exporter (after the items of freight, insurance and import duties are controlled for) would cause a discrepancy between supply and demand in the updated table. For example, trade balances from the Chinese statistical sources do not necessarily match the trade balances reported by its trading partners.¹¹ Moreover, any assumption we use in the updating, such as the assumptions on services trade, on the share of freight, insurance and import duties in total imports, may all result in discrepancies.

Nevertheless, with the exceptions of Singapore and Taiwan (R.O.C.), the residuals calculated from the GDP identity are below 5% of the GDP. In the case of Singapore, the high residual may be explained by Singapore's relatively significant re-export trade that we could not correct for given data limitations. The discrepancies in the case of Taiwan (R.O.C.) may have to do with the fact that the COMTRADE database does not contain data for Taiwan (R.O.C.), and thus we had to make some assumptions on Taiwanese trade.¹² For transparency purposes, we report a residual line when presenting the contribution ratios of final demand to value-added. This residual line stands for a part of value-added, which remains unexplained.

A second way of checking the sensitivity of the updating procedure is to update the 1995 AIO Table to year 2000 values, derive the main indicators used in our analysis, and compare the results from this updated table with the "official" 2000 AIO Table by IDE-JETRO. To do this, we calculate two measures: the Leontieff coefficients and the so-called "Contribution ratios of final demand to value added". The Leontieff coefficients from the updated 1995 AIO table indicate less significant changes in the production network between 1995 and 2000, and a stronger concentration of suppliers of inputs than the "official" 2000 AIO. The results from the impact of final demand on value added analysis are not significantly different from the

¹¹ US Department of the Treasury (2007), Report to Congress on International Economic and Exchange Rate Policies, Appendix II China's Trade Data, June 2007.

¹² Although we tried alternative estimates for Taiwanese trade assuming similarities in the trade structure of Taiwan (R.O.C.) and China and other NIE3 countries, we could not lower the residuals.

findings of the "official" 2000 AIO table, the main difference being that the updated 1995 AIO table overestimates the dependence on domestic demand by 2 percentage points and underestimates the dependence on the rest of the world by the same extent.¹³

5. ANALYSIS USING THE AIO TABLES

We use the updated 2006 AIO table in two different analyses. First, we calculate the "backward linkages" of production, which helps us to describe the inter-linkages of the emerging Asian countries in the production process. Second, as the main contribution of our paper, the reliance of each country's value added on domestic demand, intra-regional and extra-regional demand is computed. Comparing results from the 1995, 2000, and the updated 2006 AIO tables we also report the evolution of major trends in emerging Asia's production and trade dynamics.

5.1. "Backward linkages" of production

The backward linkages of production are measured by the Leontieff coefficients of the 2006 AIO table. The Leontieff coefficients of the AIO table are calculated as follows:

Let $\alpha^{ij} = \frac{A^{ij}}{X^{j}}$, where *i*=(Indonesia, Malaysia ..., US) is the supplier country, *j*=(Indonesia, Malaysia ..., US) is the demand country, A^{ij} is input from supplier country *i* used in the demand country *j*'s production, and X^{j} is total production of demand country *j*. Then the AIO table can be written in a matrix form as:

$$\begin{bmatrix} \alpha^{II} & \alpha^{IM} & . & \alpha^{IU} \\ \alpha^{MI} & \alpha^{MM} & . & \alpha^{MU} \\ . & . & . & . \\ \alpha^{UI} & \alpha^{UM} & . & \alpha^{UUI} \end{bmatrix} x \begin{bmatrix} X^{I} \\ X^{M} \\ . \\ X^{U} \end{bmatrix} + \begin{bmatrix} F^{II} \\ F^{MI} \\ . \\ F^{UI} \end{bmatrix} + \begin{bmatrix} F^{IM} \\ F^{MM} \\ . \\ F^{UM} \end{bmatrix} + \dots + \begin{bmatrix} Q^{I} \\ Q^{M} \\ . \\ Q^{U} \end{bmatrix} = \begin{bmatrix} X^{I} \\ X^{M} \\ . \\ X^{U} \end{bmatrix},$$

where F^{ij} is vector of final demand (sum of consumption and investment), Q^{i} s are export vectors to Hong Kong, the EU, and the rest of the world, respectively. For details see Appendix A.1.

The matrix notation can be written in short as: **AX+Y=X**

¹³ For more details, see Appendix A.4.

To answer the question how much production is needed to meet one unit of demand, the system of equations should be solved to *X*:

$X=(I-A)^{-1}Y=BY$, where *B* is called the Leontieff coefficient matrix.

The B^{ij} element of the matrix indicates the number of unit of production needed in country *i* (the supply country) to produce one unit of value added in country *j* (the demand country). The Leontieff coefficient matrix helps us to analyze production linkages across the countries in the region via trade of intermediate inputs.

The results of the analysis of backward linkages of production are summarized on Figure 2, which illustrates the Leontieff coefficients of the 1995, 2000 and updated 2006 AIO tables. The vertical axis of the Figure shows the supplier countries, and the horizontal axis represents the demand countries. For example, in the bottom-left corner, one can read the number of units of production needed in the United States to produce one unit of value added in Indonesia, or alternatively, the share of imports from the United States in Indonesian value added. The number of units is represented by the different colors in the Figure. According to Figure 2, in order to produce one unit of value added in Indonesia in 2006, approximately 0.03 units of production were needed in the United States. To put it differently, the import content of the Indonesian production from the United States was approximately 3%.

There are two main facts that stand out as results of the analysis. First, the dominance of horizontal formations in the Figure 2 indicates *that suppliers are highly concentrated in emerging Asia*, i.e. there are only a few countries that provide the bulk of inputs for production in the region. According to Figure 2, the main suppliers of inputs in the region are Japan, the United States, and most recently also China. As regards countries of demand, Malaysia, Singapore and Taiwan (R.O.C.) are the countries where imported inputs account for the highest share of value-added, while in the larger, less-open economies such as Korea and China, the import content of value-added is lower.

Second, *the role of major supplier is changing*. In 1995, emerging Asia used inputs mostly from Japan and from the United States in its production process. (The EU is not in the production matrix of the AIO table, i.e. it is not included in this exercise.) The pattern of production segmentation in the region was determined by the offshoring activities from these economies. By 2000, Japan and the United States still being dominant, the NIE3s and China seem to have emerged as suppliers of input material. A major change occurred by 2006, when for most countries in the region, China became a more important source of inputs than Japan and the United States.

This trend can be explained by the increasing delocalization of manufacturing production from advanced economies. For example, Toyota has created a global operating platform in recent years that operates without major Japanese inputs (Dieter (2006)). Nevertheless, the phenomenon does not come down to auto industry exclusively, but also present in other sectors, as documented by a wide range of literature (see e.g. Dieter (2006), Gauliner et al. (2005), Gangnes and Van Assche (2008) or Luthje (2004)).



Figure 2 Backward linkages of production (Leontieff coefficients)

Note: The figure depicts the Leontieff coefficients of the AIO matrices. Sources: AIO tables 1995, 2000, and authors' calculations.

ECB Working Paper Series No 993 January 2009 Figure 2 shows no evidence of clear specialization on final stage assembling in any of the countries. Final stage assembling would show up as a vertical formation in the Figure, indicating that several suppliers provide inputs for the assembler country. Although, the supply of inputs is increasingly diversified across the region due to intensifying production segmentation, no clear vertical pattern has emerged. This finding is striking in the case of China in particular, a country that has become the major export platform in the region supposedly via specializing in final stage assembling.

However, when interpreting the results one has to keep in mind that Figure 2 hides a significant heterogeneity in the data. Production linkages can differ by firms and sectors. Haddad (2007) describes various production networks within the machinery sector. In the road vehicles industry, all economies in East Asia including China, export a significant share of parts to Japan, and China also exports a large share of parts to the EU and the US. The electrical machinery sector, on the other hand, provides examples of final stage assembling. In electronics, inputs come from Japan and the NIEs and assembling and exports of final goods is done by several East Asian economies (China, Indonesia, Malaysia, and the Philippines) independently. In the electrical appliances and the computer industry East Asian economies export a high share of parts to China for final assembling and China exports the final products to the EU and the US.¹⁴

Due to data limitations, however, we are unable to update the 2000 AIO Table at the sectoral level and thus our analysis can not take into account the data heterogeneity across sectors.

5.2. Contribution ratios of final demand to value added

In the previous exercise we described production linkages via flows of intermediate inputs. Leontieff coefficients measured the production needed in the supply countries in order to produce one unit of value added in the demand country. Now, we extend the analysis further. First, we do not only take into account inputs needed for production, but also direct imports needed to meet final demand (consumption, investments and exports) in the demand country. Thus, rather than focusing on intermediate goods only, we also include flows of final goods in the analysis. Moreover, rather than gross production, the analysis focuses on value added implied in the supply country.

These changes make it possible, in the first step, to calculate the impact of final demand from demand countries on the value added of supply countries. In the next step, we can split up the

¹⁴ According to the 2000 AIO table China's role as final assembler also prevails in the textile industry.

value added of the supply countries by final demand components: by domestic demand and by final demand from other countries, i.e. exports to different destinations.¹⁵ By doing so we are able to measure the dependence of the supply countries' value added on domestic, intra-regional and extra-regional demand.

The calculations are made in two steps.

The impact of final demand on value added (IFv). The impact of final demand (from demand country j) on the value added of supply countries is calculated according to the following formula:

 $IFv^{j} = \hat{\upsilon} * B * f^{j},$

where v is a diagonal matrix consisting the elements of $v^j = V^j / X^j$ (the ratio of V value-added to total production X in the demand country), B is the Leontieff coefficient matrix, f^j is a column vector of final demand in the demand country j.¹⁶

Contribution ratios of final demand to value added. The contribution ratio (CR) of final demand from demand country j to the value added of supply country i is given by the formula:

$$CR_i^{\ j} = IFv_i^{\ j} / \sum_j IFv_i^{\ j} ,$$

where IFv_i^j stands for the *i*th row of matrix IFv^j , representing the impact of final demand from country *i* on the value-added of supply country *j*.

The main findings of the analysis of the contribution ratios are summarized in the Tables 2-4.

$$\begin{split} v^{I}B^{II}f^{II} + v^{I}B^{IM}f^{MI} + ... + v^{I}B^{IU}f^{UI} \\ IFv^{I} = & v^{M}B^{MI}f^{II} + v^{M}B^{MM}f^{MI} + ... + v^{M}B^{MM}f^{UI} \\ & \dots \\ v^{U}B^{UI}f^{II} + v^{U}B^{UM}f^{MI} + ... + v^{U}B^{UU}f^{UI} \end{split}$$

where the first row of the matrix indicates the impact of a one unit increase in Indonesian final demand on Indonesian value-added, the second row the impact on Malaysian value-added, etc. Interpreting the elements in the first row (from left to right), the Indonesian value-added is stimulated by Indonesian domestic demand to the extent that Indonesian domestic demand consumes products from domestic supply $(B^{II*}f^{II})$, plus the production of Indonesian inputs needed to produce the final goods imported from Malaysia $(B^{IM*}f^{MI})$, from the Philippines $(B^{IP*}f^{PI})$ etc.

¹⁵ Intuitively the split up of the value-added is based on the supply-demand identity: the value-added produced in the supply country is either consumed domestically, or exported to other countries.

¹⁶ For interpretation purposes, let's assume the case of a one unit increase in final demand of Indonesia (f^d) . The impact of Indonesian final demand on production of countries in the matrix (IFv^I) is the following:

	1995	2000*	2000	2006
Domestic demand	72.3%	70.3%	68.9%	64.3%
Intra-regional trade	5.3%	5.8%	5.8%	6.8%
G3	11.4%	13.8%	16.1%	15.7%
within that EU	2.1%	2.4%	4.7%	5.8%
Japan	3.9%	4.1%	4.1%	3.2%
US	5.3%	7.3%	7.3%	6.6%
RoW	11.2%	10.1%	10.1%	14.2%
residual				-1.3%

Table 2. The impact of final demand on value added, emerging Asia

Table 3. The impact of final demand on value added, China

	1995	2000*	2000	2006
Domestic demand	79.7%	79.3%	79.4%	69.7%
Intra-regional trade	2.0%	1.8%	1.8%	2.4%
G3	10.7%	12.5%	13.9%	16.6%
within that EU	2.1%	2.3%	3.8%	6.4%
Japan	4.1%	3.6%	3.6%	3.0%
US	4.5%	6.6%	6.6%	7.1%
RoW	8.0%	6.5%	6.5%	13.2%
residual				-2.4%

Table 4. The impact of final demand on value added, NIE3 and ASEAN4

	1995	2000*	2000	2006
Domestic demand	68.4%	63.0%	60.3%	57.9%
Intra-regional trade	7.0%	9.0%	9.0%	11.9%
within that China	2.1%	3.8%	3.8%	7.2%
G3	11.7%	14.9%	17.8%	14.6%
within that EU	2.1%	2.5%	5.4%	5.1%
Japan	3.9%	4.5%	4.5%	3.5%
US	5.8%	8.0%	8.0%	6.0%
RoW	12.9%	13.1%	13.1%	15.3%
residual				0.0%

Sources: AIO tables 1995, 2000, and authors' calculations.

Note: 2000* refers to EU3 under the EU line, and is directly comparable with 1995 results.

Emerging Asia consists of China, the NIE3 (Korea, Singapore and Taiwan (R.O.C.)), and the ASEAN4 (Indonesia, Malaysia, the Philippines and Thailand).

Adjusted for Hong Kong's trade, the original (not re-exported) imports of Hong Kong is taken as intra-regional demand.

The residual indicates the non-statistical discrepancy in the GDP identity of the updated 2006 AIO table (for details see Section 4.)

The Tables 2-4 present the contribution ratios of four major final demand aggregates: domestic demand, intra-regional demand (the sum of exports to emerging Asian countries),

the G3 demand (exports to United States, EU15 and Japan) and exports to the rest of the world. The contribution ratios are presented separately for the following supply countries: emerging Asia (Table 2), China (Table 3), and non-China emerging Asia (Table 4).

The tables contain two columns for 2000, which refer to different country composition of the EU. The column marked with asterisks refers to EU3 (Germany, France and the United Kingdom) data, and is thus comparable with 1995 numbers in the first column, while the other column with no asterisks refers to EU15 and comparable with 2006 data in the last column.

The following results stand out from the analysis of emerging Asia as a whole (Table 2). Approximately two-thirds of the value added of emerging Asian countries is determined by domestic demand, while the share of external demand is around one-third.¹⁷ More precisely, external factors accounted for 36.7% of the value added in 2006, implying *a significantly lower dependency of emerging Asia on exports than suggested by total trade data* (53%¹⁸). In addition, 6.8% of emerging Asia's value added was due to intra-regional demand, lowering the reliance of value added on extra-regional markets to below 30%. In 2006, the G3 countries accounted for slightly more than half of the extra-regional demand (15.7% of the value added) with the United States (6.6% of the value added) being the most important market, followed by the EU15 (5.8%) and Japan (3.2%).

The changes in the impact of the final demand components give some interesting insights. Since 1995, there is a trend increase in export dependence, indicating no sign of "decoupling", but more an increasing integration of emerging Asian countries to global trade. Dependence on intra-regional trade has also increased, in line with the strengthening of economic integration in emerging Asia. Despite its rising importance however, intra-regional trade has not compensated for the falling share of domestic demand in value added. Consequently, *the exposure of emerging Asia to extra-regional markets has increased*.

There have been differing trends in the sources of extra-regional demand in 1995-2006. The share of US demand increased between 1995 and 2000 from 5.3% to 7.3%, with a relatively stable share of the EU3 and Japan (close to 2% and 4%, respectively). After 2000, however, both the US and Japanese shares started to decline, in parallel with a significant increase in the share of EU15 from 4.7% to 5.8%. As a result, the *dependence of emerging Asia's value added on demand from the G3 economies declined slightly between 2000 and 2006*. However,

¹⁷ The share of domestic demand of total value added in emerging Asia is significantly lower than in the advanced economies. Based on the AIO, the share of domestic demand of value added in 2006 was 91% in the United States, and in Japan 86%. Given that EU15 is not included in the production matrix of the AIO, comparable statistic is not available. See table A.6. for more details.

¹⁸ Calculated as exports per GDP after aggregating National Accounts data at country level from the CEIC database.

the higher exposure of emerging Asia to extra-regional markets after 2000 was due to *stronger trade linkages with the rest of the world*, with its share in emerging Asia's value added increasing from 10.1% to 14.2% between 2000 and 2006.¹⁹

China, partly due to its size, is still less dependent on external markets than other countries in the region (Table 3). However, between 2000 and 2006, the share of external demand in the value-added of China increased substantially from 20% to 30%.

The comparison of China with the NIE3 and ASEAN4 countries by the sources of demand reveal an interesting pattern of division of labour within the region. Since 2000, China "outcrowded" the NIE3 and ASEAN4 countries from the G3 trade, and albeit its lower degree of openness, China had a higher exposure to the G3 markets in 2006 than the non-China block (16.6% vs. 14.6%, respectively). This finding is in line with the previous remark on the emergence of China as a major export platform in the region. As regards their exposure to demand from the rest of the world, China, the NIE3 and ASEAN4 countries were all more dependent on these markets in 2006 than in 2000. However, the increase in China's dependence was substantially stronger than that of the non-China block.

Finally, Table 4 also provides some information on the importance of China as a source of final demand within the region. *The dependency of the NIE3 and ASEAN4 countries' value added on Chinese markets was relatively low, at 7.2% in 2006*. The main channel of this impact, as shown by the import content analysis before, is via imported inputs to local production rather than direct imports of final goods. Albeit still at low level, *the exposure to demand from China increased substantially in recent years and almost doubled since 2000*. Moreover, by 2006 China became more important market for the NIE3 and ASEAN4 countries than the United States. More detailed country by country results are shown in Appendix A.6.

6. CONCLUSIONS

The paper contributed to the "decoupling debate" – i.e. whether the business cycle dynamics in emerging Asia have recently become less sensitive to the global demand trends - using a novel method based on an update of the Asian International Input-Output (AIO) table. In particular, the study analyzed the dependence of emerging Asia's value added through trade and production linkages on intra-regional demand, and on demand from the advanced economies, especially the United States, the EU15 and Japan. The updated 2006 AIO table

¹⁹ The reasons behind China's opening up to the rest of the world are still to be investigated. One possible explanation being that China increased its manufacturing trade surplus against the rest of the world in order to compensate its growing trade deficit in oil and raw materials in this relation.

was used in two ways. First, we calculated the "backward linkages" of production, which allowed us to describe the inter-linkages of the emerging Asian countries in the production process. Second, as the main contribution of our paper, the dependency of each country's value added on domestic demand, intra-regional and extra-regional demand was computed.

The main findings of the paper are the following. First, only about one-third of the value added in emerging Asian countries is determined by external demand, significantly lower than the 50% exposure suggested by the aggregate trade data, while domestic demand contributes around two-thirds to the value added. Second, the dependence of emerging Asia's value added on export markets has steadily risen since 1995, a phenomenon in line with increasing global trade integration, and a clear evidence against the decoupling view. Third, although intra-regional and Chinese markets have both gained importance, they still account for only around 7% of the final demand. This share is also below the one suggested by trade data.

As it is evident from these results, the paper finds no support for the decoupling view. At the same time, however, it finds that, if the bias in trade data due to the segmentation of production is accounted for, the exposure of emerging Asia is significantly lower than suggested by trade statistics. In other words, on the one hand we find no evidence of decoupling, but on the other hand we calculate that emerging Asia is less "coupled" with the rest of the world than trade data suggests.

When interpreting the results, one should note the caveat that the analysis of the real linkages with the AIO table can only capture the direct trade effects, i.e. neither any "second-round" effects of an export slowdown on domestic demand via lower employment, wages or investment, nor any financial market or policy related channels are accounted for.

REFERENCES

Asian Development Bank (2007), Uncoupling Asia: Myth and Reality, *Asian Development Outlook 2007*.

Asian Development Bank (2007), The Global Slowdown in Emerging Asia, Asian Development Outlook 2008.

Asian Development Bank (2008), *Emerging Asian Regionalism – A Partnership for Shared Prosperity*. Available at <u>http://aric.adb.org/emergingasianregionalism/</u>.

Cui L. and Syed M. (2007), The Shifting Structure of China's Trade and Production, IMF Working Paper No. 214.

Dees, S. and Vansteenkiste I. (2007), The transmission of US cyclical developments to the rest of the world, ECB Working Paper No. 798, August 2007

Dieter, H. (2007), Transnational Production Networks in the Automobile Industry, Notre Europe Studies and Research No. 58.

Gangnes B. and Van Assche A. (2008), China and the Future of Asian Electronics Trade, CIRANO, Scientific Series, 2008 Montreal.

Gaulier G., Lemoine, F. and Unal-Kesenci, D. (2006), China's Emergence and the Reorganization of trade Flows in Asia, CEPII Working Paper No 2006-05.

Gaulier G., Lemoine, F. and Unal-Kesenci, D. (2005), China's Integration in East Asia: Production Sharing, FDI and High-Tech trade, CEPII Working Paper No 2005-09.

Haddad, M. (2007), Trade integration in East Asia: The Role of China and Production Networks, World Bank Policy Research Working Paper 4160, March 2007.

Haltmaier J. T, Ahmed, S., Coulibaly, B., Knippenberg, R., Leduc S., Marazzi M. and Wilson B.A (2007), The Role of China in Asia: Engine, Conduit, or Steamroller?, Board of Governors of the Federal Reserve System, International Finance Discussion Papers No. 904.

He, D. and Zhang W. (2008), How Dependent is the Chinese Economy on Exports and in What Sense Has Its Growth Been Export-Led?, Hong Kong Monetary Authority Working Paper 14/2008.

International Monetary Fund (2007), Decoupling the Train? Chapter 4 in the *World Economy Outlook 2007*, April.

International Monetary Fund (2007), Asia and Pacific, *Regional Economic Outlook 2007*, October.

Koopman R., Wang Z. and Wei, S (2008), How Much of Chinese Exports is Really Made in China?, NBER Working Paper 14109.

Meng, B. and C. Qu (2007), Application of the Input-Output Decomposition Technique to China's Regional Economies, Institute of Developing Economies Discussion Paper No. 102, April 2007.

Mori, T. and H. Sasaki (2007), Interdependence of Production and Income in Asia–Pacific Economies: An International Input–Output Approach, Bank of Japan Working Paper Series No.07-E-26, November 2007.

Nag, B., Banerjee, S. and Chatterjee, R. (2007), Changing Features of the Automobile Industry in Asia, Asia-pacific Research and Training Network on Trade Working paper Series, No. 37, July 2007.

Luthje, B. (2004), Global Production Networks and the Industrial Upgarding in China: The Case of Electronics Contract Manufacturing, East-West Center Working Papers Economics Series No. 74.

Smakman, F. (2004), Developments in the Global Garment Industry and the role of Asia, Universiteit Utrecht, <u>http://igitur-archive.library.uu.nl/dissertations/2004-0616-130904/c1.pdf</u>

US Department of the Treasury (2007), *Report to Congress on International Economic and Exchange Rate Policies*, Appendix II Chain's Trade Data, June 2007.



Appendix A.1. The structure of the AIO table

The structure of the 2000 AIO table is shown in the Figure A.1 below. The scheme is simplified as it does not show the sectoral dimension of the table.²⁰ The way of reading the AIO table is very similar to the way of reading a standard input-output table. The three main blocks are intermediate demand (A), final demand (F) and exports (L). The first column of the production (or intermediate demand) block contains the supplier countries and the first row the use countries. As an example, A^{II} stands for the value of domestic inputs for production in Indonesia, and A^{IM} is the value of inputs from Indonesia used in the Malaysian production (Malaysian imports of intermediate goods from Indonesia) etc.

The row sums are the total outputs (Xⁱ). The elements of the rows describe the purpose the output of each country is used for. For example, the first row describes what amount of Indonesia's total output is used as input for domestic production (A^{II}), as input for production in other countries of the production matrix (A^{IM} to A^{IU}), consumed / invested domestically (F^{II}), consumed / invested by other countries of the production matrix (F^{IM} to F^{IU}), and finally exported to Hong Kong (L^{IH}), the EU15 (L^{IO}) and the rest of the world (L^{IW}).

The column sums of the production block are the total inputs, which by definition equal the total outputs ($X^{i}s$). The first column reads as follows: A^{II} indicates the domestic input content of Indonesian production, A^{MI} to A^{UI} are the input contents of Indonesian production from each country of the production matrix, A^{HI} , A^{OI} and A^{WI} are inputs from Hong Kong, the EU15 and the rest of the world, BA^{I} and DA^{I} are freight and insurance and import duties and taxes. The difference of total output and total intermediate inputs is the value-added (V^{I}).

 $^{^{20}}$ The main reason being that due to data limitations the update of the AIO matrix was possible only at a country level and thus we will focus on the aggregate / country level throughout the analysis. The numbers in the matrix are sectoral aggregates.

Figure A.1. The schematic layout of the 2000 Asian international Input-Output Table

The schematic image of the 2000 Asian international input-output table

				=	nterme	ediate	Demai	(A) br							-Inal D	eman	d (F)				EX	port (I	(-		
		sisənobnl	eizyeleM	səniqqilinq	Singapore	bneliedT	6nin3	newieT	Korea	ueder	.A.2.U	BISANODIN	serinailing	Singanore	bneliedT	6nina	newieT	Korea	neqel	.A.2.U	Export to	Export to EU	Export to R.O.W.	Discrepan Discrepan	Outouts
	code	(AI)	(MM)	(AP)	(AS)	(AT)	(AC) (AN) (AK) (<i>i</i>	4) (r	AU) (F	I) (F	VI) (FF) (FS	() (FT	(FC	(FN	(FK)	(FJ)	(FU)	(H)	(LO)	(LW)		X
Indonesia	(IAI)	Ρ	A ^{IM}	A ^{IP}	A ^{IS}	A ^{IT}	A ^{IC}	, A ^{IN}	4 IK /	1	A ^{IU} F	- 	μ	Ľ.	E F	о Ч	FIN	۲K	L L	۴ ^{וע}	Ľ_	L ^{IO}	L ^{IW}	ā	$\overline{}$
Malaysia	(MM)	P™	A ^{MM}	A ^{MP}	\mathbf{A}^{MS}	A^{MT}	A ^{MC}	A ^{MN}	A ^{MK} A	√ ^{MJ}	MU	ΈL	NM MI	۲ ۲	s F ^{MT}	¥ ⊻	MM ^L	₽ ^{MK}	ΕW	Е ^{ми}	LMH	L ^{MO}	L ^{MW}	Σ	⊻×
Philippines	(AP)	API	APM	A^{PP}	A^{PS}	A^{PT}	A^{PC}	APN	3 ^{PK} ∠	\ ^{PJ} ∉	PU F	Ē	ĕ ₽	۲ ۲	2 La	а <u>т</u> .	Nd L	ЪЧ	۲- ۲	Ε ^{ΡU}	Ч	L ^{PO}	M	٩	ځ
Singapore	(AS)	A ^{SI}	A^{SM}	A^{SP}	A^{SS}	A^{ST}	A^{SC}	A ^{SN}	a ^{sk} ∠	\ ^{SJ} ⊭	SU F	si L	M F	۲ ۲	S F ST	-SS -	E ^{sn}	F ^{sk}	FsJ	F ^{su}	L ^{SH}	L ^{so}	L ^{SW}	S	\times^{s}
Thailand	(AT)	₽⊥	A^{TM}	A^{TP}	A^{TS}	₽	A^{TC}	' A [™]	3 ^{TK} ∠	1_1 F		Ē	Ľ£	۲ <u>د</u>	EL .	Ĕ.	NTT NTT	F ^{TK}	Ē	F ^{TU}	Ę	L ^{T0}	≥	D	$\stackrel{\scriptscriptstyle \vdash}{\times}$
China	(AC)	A^{CI}	A^{CM}	A^{CP}	A^{CS}	A^{CT}	A^{cc}	A ^{CN}	a ^{ck} ∠	\cı ⊧	CU F	ц С	ы М	ъ Ч	s F ^{CT}	ос С	E ^{CN}	FcK	F _C	F ^{cu}	F	L ^{co}	L ^{CW}	о О	$\stackrel{\circ}{\times}$
Taiwan	(AN)	ANI	A ^{NM}	ANP	A^{NS}	A ^{NT}	A ^{NC}	A ^{NN}	A ^{NK} A	√ NJ	- NU	°⊾ z	MI	Z L d	s F ^{NT}	Ž.	NN L	F ^{NK}	N.	F ^{NU}	LNH	۲ N	Γ	Z	×
Korea	(AK)	A ^{KI}	A ^{KM}	A^{KP}	A^{KS}	A^{KT}	A ^{KC}	A ^{KN}	3 ^{KK} ∠	\KJ ⊱	KU F	Ť⊥ ⊽	Υ ¥L	۲ ۲	5 FKT	¥ L	E ^{KN}	F ^{KK}	Ъ ^К	F ^{ku}	LKH	L ^{KO}	L ^{KW}	Š	××
Japan	(A)	٩	A ^{JM}	A ^{JP}	A ^{JS}	Ч	A ^{JC}	, ∧″A	4 ^{JK} ∕	۲ n		ند _	ſ⊥ ⊻	۲ <u>د</u>	Ľ.	°L.	Ч Ч	ЪЧ	ΓĿ	FJU	۳J	٥ſ	۳	Ō	\sim
U.S.A.	(AU)	ΑUI	A ^{UM}	A ^{UP}	A ^{US}	A ^{UT}	A ^{uc}	A ^{UN}	A ^{UK} A	/ nn /	/un F	ъ п	ым Fu	P P	s F ^{UT}	Ъnц.	E ^{UN}	F ^{UK}	ЪШ	F ^{υυ}	L ^{UH}	Luo	Luw	ŋ	\sim
Freight and Insurance	(BF)	BA	BA^{M}	BA^{P}	BA^{S}	BA^{T}	BA^{C}	BA ^N F	3A ^K E	3A ^J B	A ^U B	F ¹ Bf	-M BF	P BF	s BF ¹	BF	BF ^N	BF ^K	ΒF ^J	ΒF ^U					
Import from Hong Kong	(CH)	A ^{HI}	A^{HM}	A^{HP}	A^{HS}	A^{HT}	\mathbf{A}^{HC}	A ^{HN}	A ^{HK} A	(HJ F	HU F	t⊥ ≖	HM F ^H	ь РН	s F ^{HT}	Н	E ^{HN}	F ^{HK}	Ε ^{ΗJ}	F ^{HU}					
Import from EU	(CO)	A ^{ol}	A ^{oM}	A ^{OP}	A ^{os}	A ^{OT}	A ^{oc}	A ^{on}	JOK A	o [,] A	on F	о Б	M Fo	0 ط	s F ^{o1}	о Ч	E ^{on}	Fok	Fol	Fou					
Import from the R.O.W.	(CW)	A ^{WI}	A ^{WM}	A^{WP}	A^{WS}	A^{WT}	AWC	A ^{WN} /	NWK A	A LW	wu F	MI F ^V	W F ^W	P F ^W	s F ^{WI}	F ^{WC}	E ^{WN}	F ^{WK}	F ^{WJ}	F ^{wu}					
Duties and Import																									
Commodity Taxes	(DT)	DA	DA^{M}	DA^{P}	DA^{S}	DA^{T}	DA^{C}	DA ^N [DA ^K E	A ^J E	A ^U D	F' DI	PF PF	PF	^S DF	ΔĽ	° DF	DF ^K	DF	DF ^U					
Value Added	(vv)	~	۸	۹>	٧ ^S	>	< ^C	۸N	< K	۔ ^	ر ا														
Total Inputs	(XX)	×	м×	٩×	×s	X^{T}	\mathbf{x}^{c}	×٧	××	_×	٦×	*	ich cell o	F A** an	d F** rep	resents	a matrix	of 76 x 7	6 and 76	ó x 4 dim	ension, re	espective	الا. ال		
la a columnities direction each cell in	the table of	off anor	v innut of	amnociti	ane of in	ductriae						Ē	ing to the	و 1 ¹ ت	dumo fro	the le	ft cide of	the table	it chow	us the co	nnocitior	of doc	de and		
of respective country. A for example	shows the i	nput cor	npositior	rs of Ind	onesian	industrie	s					Serv	ices that	have go	ne to fina	al demar	nd sector	s of Indoi	nesia. F ^{II}	and F ^M ,	for exam	u yu u Iple, maj	ps the		
vis-à-vis domestically produced goods	and service	s, i.e. dc	mestic t	ransactio	ons of In	donesia.						the	inflow int	o Indon	esian fina	l demar	id sectors	s, of good	ls and se	irvices do	mesticall	y produc	ced and		

 a^{M} in contrast shows the input composition of Indonesian industries for the imported goods and services from Malaysia. The cells $a^{P_1}, a^{P_1}, a^{P_1}, a^{P_1}, a^{P_1}, a^{P_1}, a^{P_1}, a^{P_1}$

allow the same interpretation for the imports from other countries. BA and DA give international freight & insurance and taxes on these import transactions.

of those imported from Malaysia, respectively. The rest of the column is read in the same manner as is done for the 1st column of the table. L⁴¹, L^{*0}, L^{*W} are exports (vectors) to Hong Kong, EU and the Rest of the World, repectively. Vs and Xs are value added and total input/output, as seen in the conventional national I-O table.

Appendix A.2. The updating procedure

The scheme of the updating procedure is shown in Figure A.2.1. The updating procedure and the data sources used are similar to Mori and Sasaki (2006), with two main improvements. First, the trade data used for the update differs by type of goods, and thus takes into account the shift in the composition of trade from final to intermediate goods. Second, the data are adjusted for Hong Kong's entrepot trade.²¹

The starting point of the updating procedure is the 2000 AIO table. In general, the 2006 value of a specific cell in the AIO table is calculated by multiplying the 2000 value of the cell by its nominal growth rate in 2000-2006. The steps of the procedure and the estimation of the 2006 / 2000 growth rates are as follows:

Intermediate demand block (A)

- Value added (V_{t+1}^{j}) . The value added growth rates for each country are taken from National Account statistics (the datasource being the CEIC database).
- Total output (X^j_{t+1}). With the exception of the United States, direct information on total economy's output is not available. Thus, total (gross) output is estimated by applying the output / value added ratio in the manufacturing sector to the total economy's value added. Data on manufacturing value added is from national accounts sources, output data are from industrial statistics (datasource CEIC).
- Imported inputs (A^{ij}_{t+1}). The calculation of growth rates of imported inputs draws on two data sources. First, in order to keep consistency, we use the growth rate of imports from the National Accounts statistics. The advantage of using this statistics is that it includes trade of goods and services, while the disadvantage is that it does not provide information by the direction of trade. To get an estimate for changes in the direction of trade, we combined National Accounts import growth with the information from the COMTRADE. The COMTRADE database provides information on imports not only by direction, but also by type of good, i.e. it helps us to take into account the increasing share of intermediate inputs in total imports. (The classification of imports by type of goods I described in Appendix A.3.)

²¹ In 2006, 95% of the Hong Kong's exports were re-exported. This re-exports are overwhelmingly originated from China and aimed at overseas markets. As Hong Kong is considered as part of the emerging Asian region, then if not corrected for, the Hong Kong re-exports may result in an overestimation of the intra-regional and underestimation of extra-regional demand. Given that Hong Kong is not included in the production matrix of the AIO table the adjustment has to be made as an additional exercise.

However, the COMTRADE only includes data on goods trade, thus we have to assume that changes in imports by direction and by type of good are similar for goods and services.²²

The formula used to estimate the imported input growth rate is the following:

$$(\operatorname{int} M_{t+1}^{*ij} / \operatorname{int} M_{t}^{*ij}) = (M_{t+1}^{NA} / M_{t}^{NA}) * \frac{(\operatorname{int} M_{t+1}^{COMij} / \operatorname{int} M_{t}^{COMij})}{(M_{t+1}^{COM} / M_{t}^{COM})},$$

where the superscripts NA and COM stand for National Accounts and COMTRADE respectively and *intM* indicates imports of intermediate goods and *M* is for total imports.²³

- Freight and insurance and import duties $(BA_{t+1}^{j}, DA_{t+1}^{j})$. The growth rates of items freights and insurance and import duties are chosen to be the same as the import growth rate from the National Accounts. This implies the assumption of unchanged share of these items in total imports from 2000 to 2006.
- Domestic input of production (A^{jj}_{t+1}) . The domestic input content of production is calculated as a residual of the column, i.e. total inputs minus total imported inputs $(=X^{j}_{t+1}-V^{j}_{t+1}-BA^{j}_{t+1}-DA^{j}_{t+1}-DA^{j}_{t+1}-\sum A^{ij}_{t+1})$

The update of the *final demand block* (F) follows the same pattern. The final demand of each country is calculated by updating the components of consumption and investments separately.

- Total consumption and investments $(C_{t+1}^{j}, I_{t+1}^{j})$. The growth rates of consumption and investments are taken from the National Accounts statistics. Consumption is defined as the sum of private and government consumption, while investments equal gross capital formation (gross fixed capital formation plus inventories).
- Imported final goods (cF^{ij}_{t+1}) and imported capital goods (iF^{ij}_{t+1}) . The growth rates are calculated according to the formula given for the imported inputs above, with the difference that the COMTRADE data on final and capital goods are used instead of the data on intermediate goods.

$$(cM_{t+1}^{*ij}/cM_t^{*ij}) = (M_{t+1}^{NA}/M_t^{NA}) * \frac{(cM_{t+1}^{COMij}/cM_t^{COMij})}{(M_{t+1}^{COM}/M_t^{COM})},$$

$$(capM_{t+1}^{*ij} / capM_{t}^{*ij}) = (M_{t+1}^{NA} / M_{t}^{NA}) * \frac{(capM_{t+1}^{COMij} / capM_{t}^{COMij})}{(M_{t+1}^{COM} / M_{t}^{COM})},$$

²² COMTRADE data are not available for Taiwan. Import growth rate for Taiwan is taken from the National Accounts statistics, i.e. it lacks heterogeneity by countries of origin and types of goods.

 $^{^{23}}$ Note that, if the growth rate of total goods imports from COMTRADE (M^{COM}) would equal the growth rate of total goods and services imports from National Accounts (M^{NA}) then the growth rate would be simply the growth rate of intermediate goods imports.

where the superscripts *NA* and *COM* stands for National Accounts and the COMTRADE, respectively and *cM* and *capM* indicates imports of consumption and capital goods and *M* total imports.

- Freight and insurance, and import duties $(cBF_{t+1}^{j}, iBF_{t+1}^{j}, cDF_{t+1}^{j}, iDF_{t+1}^{j})$. Similar to the intermediate demand block.
- Domestically produced final and capital goods $(cF^{ij}_{t+1}, iF^{ij}_{t+1})$. Residual similar to the intermediate demand block.

Export block (L)

• Exports to Hong Kong, EU15 and RoW (L^{iH}_{t+1} , L^{iO}_{t+1} , L^{iW}_{t+1}). Growth rates are calculated in a similar manner as before, i.e. as a combination of the National Accounts' export growth rates and the COMTRADE export growth rates by country of destination (HK, EU15, RoW).

$$(EX_{t+1}^{*ij} / EX_{t}^{*ij}) = (EX_{t+1}^{NA} / EX_{t}^{NA}) * \frac{(EX_{t+1}^{COMij} / EX_{t}^{COMij})}{(EX_{t+1}^{COM} / EX_{t}^{COM})}$$

 Statistical discrepancy (Qⁱ_{t+1}). To calculate the discrepancy, the data for 2006 data are taken from the National Accounts.

Adjusting for Hong Kong's entrepot trade

According to Hong Kong trade statistics, re-exports accounted for 95% of the country's exports in 2006. Re-exports consists of goods that pass through Hong Kong without having undergone "a manufacturing process which has changed permanently the shape and nature, form or utility of the product"²⁴. Hong Kong's re-exports are overwhelmingly related to trade between China and the overseas markets. In total re-exports of Hong Kong, the share of China's exports to out-of-region markets was 39% in 2006, while the share of imports of China from out-of-region markets was 22% (Figure A.2.2). Thus, if not corrected for, the Hong Kong re-exports may result in an overestimation of the intra-regional and underestimation of extra-regional demand.

The Hong Kong trade statistics provides information on re-exports by country of origin, destination, and also by type of good and destination. Based on this information, we used the following formulas to adjust the imports of intermediate, consumer and capital goods respectively for the Hong Kong entrepot trade.

²⁴ US Department of the Treasury (2007), Report to Congress on International Economic and Exchange Rate Policies, Appendix II Chain's Trade Data, June 2007.

 $A^{ij}_{HK} = A^{ij} + \eta^{j}_{int} * \gamma^{ij} * LH^{i}$ $cF^{ij}_{HK} = cF^{ij} + \eta^{j}_{cons} * \gamma^{ij} * LH^{i}$ $iF^{ij}_{HK} = iF^{ij} + \eta^{j}_{cap} * \gamma^{ij} * LH^{i}$

where LH^{i} is exports from (origin) country *i* to Hong Kong,

 $\gamma^{ij} = \frac{RX^{*ij}}{M^{*i}}$ is the share of re-exports (RX) from (origin) country *i* to (destination) country *j* in total

imports of Hong Kong from (origin) country i, and

 $\eta_g^j = \frac{RX_g^{*j}}{RX_g^{*j}}$, is the share of intermediate, capital and consumption goods (g={int, cons, cap} type of

good) in re-exports to (destination) country j.²⁵

Figure A.2.2: The value of China-related







As a result of the adjustment, the new trade weights of China indicate a significantly higher share for the United States, the EU15 and rest of the world markets, while intra-regional markets and Japan gain relatively less in importance (Figure A.2.3). In fact, the share of the United States in China's total exports rises from 18% to 22% in 2006, the share of EU15 from 15% to 19% and the rest of the world from 36% to 39%, an almost 4 percentage point increase on average. The adjustment's impact on the shares of emerging Asia and Japan are lower, a 1.4 percentage point on average.

Source: CEIC.

Source: AIO updated.

²⁵ Given that data on re-exports by type of good is only available in relation to destination countries, we apply the assumption that the distributions across the types of goods are similar regardless of the country of origin of re-exports.

			2			ן ב ת	ש	Š N					5	a		יי	n c	וו	ra n	ט			-	-	
	-	ntermed.	iate D	eman	(A)									Final D	emanc	(L)					Expo	rt (L)			
) ode	A) Indonesia A) Malaysia	səniqqilinq A	AS Singapore	PueliedT A	enin) A	newieT (NA)	AK) Korea	neqel (S	.A.2.U €	sizənobnl (^H	eisyeleM $\widetilde{\mathbb{S}}$	səniqqilinq 开	E Singapore	bneliedT (F	Enina ()	newieT ()	F Korea	([((.A.2.U 🖯	E Export to	() EU Export to	S R.O.W. Statistical	S Discrepancy	SindinO
											C I														
Indonesia	(IV)	A''		-					;;		cF ^{II} ¡F ^{II}			=	-	-					E.		0	3	-× :
Malaysia Philippines	(AP)	A ^{MI}	A ^u t+,	$_{t}^{1} = X_{t}^{1}$	≥_1	+1-DA	/1+1-B	A ¹⁺¹⁻	ΣA^{1}_{t+1}		cF ^{PI} iF ^{PI} cF ^{PI} iF ^{PI}			CF ^u _{t+1} iF ⁱⁱ _{t+1}	=C' _{t+1} - = ^t +1-ii	cDA ^I t+ DA ^I t+1-	. ₁ -cBA' iBA ^j ₊₁	t ₊₁ -ΣcF -ΣiF ^{ij}			L H L		00	ع م	[™] × ⊀
Singapore	(AS)	A ^{sı}						/			cF ^{SI} iF ^{SI}					Į	$\left \right $	/		_	L ^{SH}	•	0	s	X ^s
Thailand	(AT)	A ^{⊤I}				*					cF ^{TI} iF ^{TI}						<u> </u>	/			E_		0		×
China	(AC)	A ^{cl}							/		cF ^{cl} iF ^{cl}						•			_	ы		0	υ	×c
Taiwan	(AN)	⊀ ⊿¤	A^{ij}_{t+1}	= A	t * (ji	ntM*	ij / 1	intM	(^t i*		CF ^{NI} iF ^{NI}	cF ^{IJ} +1	= CF ^{IJ}	* (cM	* ^{ij} t ₊₁ /	cM* ^{ij}			/	_	E ^N		a	z	××
Korea	(AK)	A ^{KI}									cF ^{KI} iF ^{KI} J	iF ^{ij} t+1	_ ا آ آ	* (cap	M* ^{!j}	1 / cap	([†] *M			_	۲¥		0	~	×
Japan	(AJ)	A ^{JI}									cF ^{JI} iF ^{JI}									,	L ۲		0	-	۲×
U.S.A.	(AU)	A ^{UI}									cF ^{UI} iF ^{UI}										L ^{UH}		0	n	Х ^U
-reight and Insurance	(BF)	3A ¹	BA ^j ₊	-1 - -	3A ¹ *	(intl\	۱* ^j	/ int/	(¹ +۱)		cBF ^I iBF ^I	cBF ^j +	1 = CB	Б ^j * (consM	*j _{t+1} /	cons	۱* ^j)							
mport from Hong Konç	(H)	A ^{HI}									cF ^{HI} iF ^{HI}	cF ^{IJ} +1	$= cF^{ij}$	* (cor	^{įį} *Msr	t+1 / C(*Msnc	(ț			$\int_{t+1}^{j} = L$	ј, * (Е	X^{*ij}_{t+1}	/ EX* ^{ij}	3
mport from EU	(CO)	4 ⁰¹	A^{ij}_{t+1}	= A	i * (i	ntM*	ij t+1 /	intM	(^j		cF ^{OI} iF ^{OI}	iF ^{ij}	= iF ^{ij}	* (cap	M* ^{ij} _t+.	1 / cap	([†] , M			0	$\int_{t+1}^{1} = 0$	D ^{ij} * (Δ* ^{ij} t+1	/ Q* ^{ij} ,)	
Import from the R.O.W (CW)	4 WI									cF ^{wl} iF ^{wl}														
Duties and Import	Í	'AC	$DA^{J}_{t_{J}}$	-1 = [⊃Ajt '	* (intl	M^{*j}_{t+1}	1 / int	tM* ^j ,		cDF ^I iDF ^I	cDF ¹ t+	$1^{-1} = c$	DF ¹ t*(conslv	۱* ^j +1 /	cons!	(^ب *ار)							
commodity laxes	(10)		-	ľ			ľ					iDF ¹		× +	apM*	_{t+1} / C	apM ^{*,}								
Value Added	$\widehat{\mathbf{N}}$	~	V_{t+1}^{j}	= V ^j	× >	*j t+1	/ V* ^j ,				 C	C ⁱ +1 =	C ⁱ *	(C* ^j +1	/ C* ^j)	_	j t+1 =	اً * (ا*	t+1 / 1*	(^t					
Total Inputs	(XX)	-×	X_{t+1}^{j}	$= X_{t}^{j}$	* (X*	* j / 1 /	X^{*j}_{t}																		
$X^{*j}_{+} = Gross productio$	n of co	untry j (N	ationa	I Accol	unts dé	ata) or	' indus	itrial g	ross out	tput	intM* ^j = inter	mediate	i spood	mports	of coun	trv j (C	OMTRA	DE +Nat	ional Ac	counts					
•								•			• .		,			•									
$V^{*J_{+}} = GDP$ of country	j (Nati	onal Accou	unts)								intM ^{*IJ} = inter	mediate	goods	imports	of cour	ntry j fr	om coui	ntry i (C	OMTRA	DE+Na	tional Ac	counts)			

Indating the 2000 Acian international innut outbut table

Figure A.2.1 Schematic chart of the updating procedure

 $consM^{*ij} = consumption goods imports of country j from country i (COMTRADE+National Accounts)$ $\label{eq:capM*} capM*^{i}{}_{t} = capital goods imports of country j (COMTRADE+National Accounts) \\ capM*^{ij}{}_{t} = capital goods imports of country j from country i (COMTRADE+National Accounts) \\ \end{cases}$

 $consM^{*j}t = consumption goods imports of country j (COMTRADE+National Accounts)$

$$\begin{split} \mathbb{C}^{\star j_{1}} &= total consumption of country j (National Accounts) \\ \mathbb{I}^{\star j_{1}} &= capital accumulation of country j (National Accounts) \\ \mathbb{E}X^{\star j_{1}} &= exports of country j to country i (COMTRADE+National Accounts) \end{split}$$

33

Appendix A.3. Classification of goods by Broad Economic Categories (BEC)

1 Capital goods	
Sum of categories:	
41	Capital goods (except transport equipment)
521	Transport equipment, industrial
2 Intermediate goods	
Sum of categories:	
111	Food and beverages, primary, mainly for industry
121	Food and beverages, processed, mainly for industry
21	Industrial supplies not elsewhere specified, primary
22	Industrial supplies not elsewhere specified, processed
31	Fuels and lubricants, primary
322	Fuels and lubricants, processed (other than motor spirit)
42	Parts and accessories of capital goods (except transport equipment)
53	Parts and accessories of transport equipment
3 Consumption goods	
Sum of categories:	
112	Food and beverages, primary, mainly for household consumption
122	Food and beverages, processed, mainly for household consumption
522	Transport equipment, non-industrial
61	Consumer goods not elsewhere specified, durable
62	Consumer goods not elsewhere specified, semi-durable
63	Consumer goods not elsewhere specified, non-durable
1 Other goods	
Sum of actogories:	
Sum of calegomes.	Motor opirit
21	

- 51 Passenger motor cars
- 7 Goods not elsewhere specified

Note: When the breakdown of goods to capital, intermediate and consumption goods is used in the paper, the three first categories are used. However, in the cases where total trade is used, then the data also includes the fourth category of Other goods.

Appendix A.4. Sensitivity analysis - results from the updated 1995 AIO table

As a sensitivity check of the updating procedure we also updated the 1995 AIO table to year 2000 and compared the results from this updated table with that from the "official" 2000 AIO table.

The updating procedure was similar to the one used in the paper: we took the 1995 AIO Table as a starting point and updated each cell according to the steps described in section A.2. Data limitations were somewhat more severe than in the original exercise. First, as no COMTRADE import data was reported by the Philippines for 1995, we used the export data reported by partner countries adjusted for the freight and insurance and import duties component. Second, Indonesian producer prices were proxied by the average producer price inflation in Malaysia, the Philippines and Thailand. Similar to the original exercise we lack COMTRADE trade data for Taiwan, applying the output / value added ratio in the manufacturing sector to the total economy's value added and assume changes in imports by direction and by type are similar for goods and services.

To test the sensitivity of our findings to the updating procedure, we calculated the Leontieff coefficients and the impacts of final demand on value added from the updated 1995 table and compared the results with the ones from the "official" 2000 AIO table.

The Leontieff coefficients from the updated 1995 AIO table indicate less significant changes in the production network between 1995 and 2000, and a stronger concentration of suppliers of inputs than the "official" 2000 AIO.







However, the results from the impact of final demand on value added analysis with the updated 1995 AIO table are not significantly different from the findings of the "official" 2000 AIO table, the main difference being that the updated 1995 AIO table overestimates the dependence on domestic demand by 2 percentage points and underestimates the dependence on the rest of the world by the same amount.

	updated 1995 AIO Table	official 2000 AIO Table	difference in pps	
Domestic demand	70%	68%	2%	
Intra-regional trade	8%	7%	0%	
G3	13%	13%	0%	
within that EU3	2%	2%	0%	
Japan	4%	4%	0%	
US	7%	7%	0%	
RoW	10%	12%	-2%	
residual	0%	-1%	1%	

Table A4.1. The impact of final demand on value added, not adjusted for Hong Kong trade



Appendix A.5. The derivation and interpretation of the main indicators used in the analysis

Backward linkages (Leontieff coefficients)

Let
$$\alpha^{ij} = \frac{A^{ij}}{X^{j}}$$

where index i=(I, M ..., U) depicts the supplier country and j = (I, M ..., U) is the country of demand. X is total production, A is intermediate inputs from the supplier country used in the production of the country of demand.

Then the AIO Table can be written as:

$$\begin{split} \alpha^{II} X^{I} + \alpha^{IM} X^{M} + ... + \alpha^{IU} X^{U} + F^{II} + F^{IM} + ... + Q^{I} = X^{I} \\ \alpha^{MI} X^{I} + \alpha^{MM} X^{M} + ... + \alpha^{MU} X^{U} + F^{MI} + F^{MM} + ... + Q^{M} = X^{M} \\ ... \\ \alpha^{UI} X^{I} + \alpha^{UM} X^{M} + ... + \alpha^{UU} X^{U} + F^{UI} + F^{UM} + ... + Q^{U} = X^{U} \end{split}$$

or in a matrix form:

$$\begin{bmatrix} \alpha^{II} & \alpha^{IM} & \cdot & \alpha^{IU} \\ \alpha^{MI} & \alpha^{MM} & \cdot & \alpha^{MU} \\ \cdot & \cdot & \cdot & \cdot \\ \alpha^{UI} & \alpha^{UM} & \cdot & \alpha^{UUI} \end{bmatrix} x \begin{bmatrix} X^{I} \\ X^{M} \\ \cdot \\ X^{U} \end{bmatrix} + \begin{bmatrix} F^{II} \\ F^{MI} \\ \cdot \\ F^{UI} \end{bmatrix} + \begin{bmatrix} F^{IM} \\ F^{MM} \\ \cdot \\ F^{UM} \end{bmatrix} + \dots + \begin{bmatrix} Q^{I} \\ Q^{M} \\ \cdot \\ Q^{U} \end{bmatrix} = \begin{bmatrix} X^{I} \\ X^{M} \\ \cdot \\ X^{U} \end{bmatrix}$$

in short: AX+Y=X

To answer the question how much production is needed to meet 1 unit of demand the system of equation should be solved to X:

$\mathbf{X} = (\mathbf{I} - \mathbf{A})^{-1} \mathbf{Y} = \mathbf{B} \mathbf{Y},$

where *B* is called the *Leontieff coefficient matrix*.

The B^{ij} element of the matrix indicates the number of unit of production needed in country i to produce 1 unit of value added in country j.

The impact of final demand on value-added

The formula used is

$$IFv^{j} = \widehat{\upsilon} * B * f^{j},$$

where *B* is the Leontieff coefficient, f^{j} is final demand of country *j* and *v* is a diagonal matrix constructed from the elements of $v^{j} = V^{j} / X^{j}$ and f^{j} is the demand vector of country *j*.

For interpretation purposes let's assume the case of a one unit increase in final demand of Indonesia (f^d) . The impact of Indonesian final demand on production of countries in the matrix (IF^I) is the following:

$$IF^{I} = \begin{array}{c} B^{II} f^{II} + B^{IM} f^{MI} + \dots + B^{IU} f^{UI} \\ B^{MI} f^{II} + B^{MM} f^{MI} + \dots + B^{MM} f^{UI} , \\ \dots \\ B^{UI} f^{II} + B^{UM} f^{MI} + \dots + B^{UU} f^{UI} \end{array}$$

where the first row of the matrix indicates the impact on Indonesian production, the second row the impact on the Malaysian production etc.

Interpreting the elements of the formula in the first row, the Indonesian production is stimulated by Indonesian domestic demand to the extent that Indonesian domestic demand consumes products from domestic supply $(B^{II}*f^{II})$, plus the production of Indonesian inputs needed to produce the final goods imported from Malaysia $(B^{IM}*f^{MI})$, the Philippines $(B^{IP}*f^{PI})$ etc.

Multiplying the IF^{i} matrix with the *v* diagonal matrix, the formula gives the level of induced value added rather than that of the induced production.

نه	
.s	
.	
er	
X	
e)	
9	
ē	
P	
Ę	
~	
ē	
1	
g	
g	
•	
p	
E	
19	
H	
e	
9	
l	
Πį	
Æ	
Ξ	
5	
+	
ິ	
8	
Ξ	
Ξ.	
5	
Ā	
Ŧ	
п	
3	
E.	
4	
2	
5	
تە	
1	
<u>d</u>	
le	
ai	
ž	
ĕ	
Ξ	
9	
-	
\triangleleft	
Ö	
<u>د</u> ه	
ā.	
Ξ	
Ц,	
~	

Year 2006 with adjustment for Hong Kong entrepot trade

	Demand countri	ies											
	Indonesia	Malaysia	Phillippines	Singapore	Thailand	China	Taiwan	Korea	Japan	NS	НК	EU15	RoW
Source countries													
Indonesia	74%	1%	%0	%0	1%	2%	1%	1%	4%	3%	0%	3%	6%
Malaysia	1%	31%	1%	2%	2%	6%	2%	2%	5%	12%	1%	6%	20%
Philippines	%0	1%	55%	%0	1%	8%	1%	1%	4%	6%	1%	6%	7%
Singapore	1%	2%	1%	26%	1%	6%	1%	2%	2%	8%	4%	8%	32%
Thailand	1%	1%	0%0	1%	54%	5%	1%	1%	5%	7%	1%	5%	21%
China	%0	%0	0%0	%0	0%	20%	0%	1%	3%	7%	%0	6%	13%
Taiwan	0%0	1%	0%0	%0	1%	16%	49%	1%	5%	12%	1%	8%	17%
Korea	%0	%0	0%0	%0	0%0	6%	1%	65%	2%	4%	1%	4%	14%
Japan	%0	%0	0%0	%0	0%	2%	1%	1%	84%	3%	%0	2%	5%
NS	%0	%0	0%0	%0	0%	%0	%0	0%	0%	91%	%0	1%	6%
The percentages indi	cate the ratio	of demand	from each cou	untry to the t	otal value-ad	dded produ	ced in the s	ource coun	tries				
The rows should sum	up to 100%,	the differen	ce being the s	um of the re.	sidual and th	ie statistico	il discrepan	cy (not incl	uded in the	table)			



European Central Bank Working Paper Series

For a complete list of Working Papers published by the ECB, please visit the ECB's website (http://www.ecb.europa.eu).

- 944 "The New Area-Wide Model of the euro area: a micro-founded open-economy model for forecasting and policy analysis" by K. Christoffel, G. Coenen and A. Warne, October 2008.
- 945 "Wage and price dynamics in Portugal" by C. Robalo Marques, October 2008.
- 946 "Macroeconomic adjustment to monetary union" by G. Fagan and V. Gaspar, October 2008.
- 947 "Foreign-currency bonds: currency choice and the role of uncovered and covered interest parity" by M. M. Habib and M. Joy, October 2008.
- 948 "Clustering techniques applied to outlier detection of financial market series using a moving window filtering algorithm" by J. M. Puigvert Gutiérrez and J. Fortiana Gregori, October 2008.
- 949 "Short-term forecasts of euro area GDP growth" by E. Angelini, G. Camba-Méndez, D. Giannone, L. Reichlin and G. Rünstler, October 2008.
- 950 "Is forecasting with large models informative? Assessing the role of judgement in macroeconomic forecasts" by R. Mestre and P. McAdam, October 2008.
- 951 "Exchange rate pass-through in the global economy: the role of emerging market economies" by M. Bussière and T. Peltonen, October 2008.
- 952 "How successful is the G7 in managing exchange rates?" by M. Fratzscher, October 2008.
- 953 "Estimating and forecasting the euro area monthly national accounts from a dynamic factor model" by E. Angelini, M. Bańbura and G. Rünstler, October 2008.
- 954 "Fiscal policy responsiveness, persistence and discretion" by A. Afonso, L. Agnello and D. Furceri, October 2008.
- 955 "Monetary policy and stock market boom-bust cycles" by L. Christiano, C. Ilut, R. Motto and M. Rostagno, October 2008.
- 956 "The political economy under monetary union: has the euro made a difference?" by M. Fratzscher and L. Stracca, November 2008.
- 957 "Modeling autoregressive conditional skewness and kurtosis with multi-quantile CAViaR" by H. White, T.-H. Kim, and S. Manganelli, November 2008.
- 958 "Oil exporters: in search of an external anchor" by M. M. Habib and J. Stráský, November 2008.
- 959 "What drives U.S. current account fluctuations?" by A. Barnett and R. Straub, November 2008.
- 960 "On implications of micro price data for macro models" by B. Maćkowiak and F. Smets, November 2008.
- 961 "Budgetary and external imbalances relationship: a panel data diagnostic" by A. Afonso and C. Rault, November 2008.
- 962 "Optimal monetary policy and the transmission of oil-supply shocks to the euro area under rational expectations" by S. Adjemian and M. Darracq Pariès, November 2008.

- 963 "Public and private sector wages: co-movement and causality" by A. Lamo, J. J. Pérez and L. Schuknecht, November 2008.
- 964 "Do firms provide wage insurance against shocks? Evidence from Hungary" by G. Kátay, November 2008.
- 965 "IMF lending and geopolitics" by J. Reynaud and J. Vauday, November 2008.
- 966 "Large Bayesian VARs" by M. Bańbura, D. Giannone and L. Reichlin, November 2008.
- 967 "Central bank misperceptions and the role of money in interest rate rules" by V. Wieland and G. W. Beck, November 2008.
- 968 "A value at risk analysis of credit default swaps" by B. Raunig and M. Scheicher, November 2008.
- 969 "Comparing and evaluating Bayesian predictive distributions of asset returns" by J. Geweke and G. Amisano, November 2008.
- 970 "Responses to monetary policy shocks in the east and west of Europe" by M. Jarociński, November 2008.
- 971 "Interactions between private and public sector wages" by A. Afonso and P. Gomes, November 2008.
- 972 "Monetary policy and housing prices in an estimated DSGE for the US and the euro area" by M. Darracq Pariès and A. Notarpietro, November 2008.
- 973 "Do China and oil exporters influence major currency configurations?" by M. Fratzscher and A. Mehl, December 2008.
- 974 "Institutional features of wage bargaining in 23 European countries, the US and Japan" by P. Du Caju, E. Gautier, D. Momferatou and M. Ward-Warmedinger, December 2008.
- 975 "Early estimates of euro area real GDP growth: a bottom up approach from the production side" by E. Hahn and F. Skudelny, December 2008.
- 976 "The term structure of interest rates across frequencies" by K. Assenmacher-Wesche and S. Gerlach, December 2008.
- 977 "Predictions of short-term rates and the expectations hypothesis of the term structure of interest rates" by M. Guidolin and D. L. Thornton, December 2008.
- 978 "Measuring monetary policy expectations from financial market instruments" by M. Joyce, J. Relleen and S. Sorensen, December 2008.
- 979 "Futures contract rates as monetary policy forecasts" by G. Ferrero and A. Nobili, December 2008.
- 980 "Extracting market expectations from yield curves augmented by money market interest rates: the case of Japan" by T. Nagano and N. Baba, December 2008.
- 981 "Why the effective price for money exceeds the policy rate in the ECB tenders?" by T. Välimäki, December 2008.
- 982 "Modelling short-term interest rate spreads in the euro money market" by N. Cassola and C. Morana, December 2008.
- 983 "What explains the spread between the euro overnight rate and the ECB's policy rate?" by T. Linzert and S. Schmidt, December 2008.

- 984 "The daily and policy-relevant liquidity effects" by D. L. Thornton, December 2008.
- 985 "Portuguese banks in the euro area market for daily funds" by L. Farinha and V. Gaspar, December 2008.
- 986 "The topology of the federal funds market" by M. L. Bech and E. Atalay, December 2008.
- 987 "Probability of informed trading on the euro overnight market rate: an update" by J. Idier and S. Nardelli, December 2008.
- 988 "The interday and intraday patterns of the overnight market: evidence from an electronic platform" by R. Beaupain and A. Durré, December 2008.
- 989 "Modelling loans to non-financial corporations in the euro area" by C. Kok Sørensen, D. Marqués Ibáñez and C. Rossi, January 2009.
- 990 "Fiscal policy, housing and stock prices" by A. Afonso and R. M. Sousa, January 2009.
- 991 "The macroeconomic effects of fiscal policy" by A. Afonso and R. M. Sousa, January 2009.
- 992 "FDI and productivity convergence in central and eastern Europe: an industry-level investigation" by M. Bijsterbosch and M. Kolasa, January 2009.
- 993 "Has emerging Asia decoupled? An analysis of production and trade linkages using the Asian international input-output table" by G. Pula and T. A. Peltonen, January 2009.